CS 302, Section 20

T.A. Andrew Baptist

http://www.cs.wisc.edu/~abaptist/cs302

TA info:

Instructor: Andrew Baptist Office: 1307 Computer Science and Statistics (1st floor) Phone: 262-6602 Office Hours: To be determined Email: abaptist@cs.wisc.edu Section info:

Section 20 meets at 6:30 MW in 1221 Comp. Sci Text: Java Software Solutions, John Lewis & William Loftus Resources:

NT Labs: CS1350, CS1351 and CS1363 Software: Microsoft Windows NT, Sun Java Workshop 2.0, Netscape Web Browser Email List: cs302-20list@cs.wisc.edu Policy info:

Late programs are not excepted under any circumstances without prior approval Programs are run through a screening program for similarities Don’t share code Grading:

Programs  (6 programs)         40%
Quizzes   (6 quizzes-drop 1)   15%
Midterm                        20%
Final                          25%